Author’s response to reviews

Title: The effectiveness of the Who's Challenging Who support staff training intervention to improve attitudes and empathy towards adults with intellectual disability and challenging behaviours: study protocol for a cluster randomised controlled trial

Authors:

Elizabeth Randell (randelle@cardiff.ac.uk)
Richard Hastings (R.Hastings@warwick.ac.uk)
Rachel McNamara (McNamara@cardiff.ac.uk)
Roseanne Knight (R.E.Knight@warwick.ac.uk)
David Gillespie (GillespieD1@cardiff.ac.uk)
Zachery Taylor (Zac.Taylor@mencap.org.uk)

Version: 2 Date: 29 Aug 2017

Author’s response to reviews:

Please see the responses to the reviewers additional comments in the attached 'WCW Protocol paper reviewer responses'